Author's response to reviews

Title: Prognostic significance of multidrug-resistance protein (MDR-1) in renal clear cell carcinomas: A five year follow-up analysis

Authors:

Chiara Mignogna (chiara.mignogna@libero.it)
Stefania Staibano (staibano@unina.it)
Vincenzo Altieri (altieri@unina.it)
Gaetano De Rosa (gaderosa@unina.it)
Giuseppe Pannone (giuseppegannone@libero.it)
Angela Santoro (angelasantoro1@hotmail.it)
Rosanna Zamparese (rosannazamparese@libero.it)
Massimino D'Armiento (giuseppegannone@virgilio.it)
Romualdo Rocchetti (vangieson@libero.it)
Ernesto Mezza (mezza@unina.it)
Mario Nasti (mario.nasti@tin.it)
Viviana Strazzullo (graziafanara@libero.it)
Vittorino Montanaro (vincenzo.santoro0@alice.it)
Massimo Mascolo (mmascolo@libero.it)
Pantaleo Bufo (p.bufo@unifg.it)

Version: 6 Date: 17 December 2006

Author's response to reviews:

Author's response to the Biomed Central Editorial Production Team

Definitive Title: 2306172411155395 Prognostic significance of multidrug-resistance protein (MDR-1) in renal clear cell carcinomas: A five year follow-up analysis

Authors:

Chiara Mignogna (chiara.mignogna@libero.it)
Stefania Staibano (staibano@unina.it)
Vincenzo Altieri (altieri@unina.it)
Gaetano De Rosa (gaderosa@unina.it)
Giuseppe Pannone (giuseppegannone@virgilio.it)
Angela Santoro (angelasantoro1@hotmail.it)
Rosanna Zamparese (rosannazamparese@libero.it)
Massimino D'Armiento (giuseppegannone@virgilio.it)
Romualdo Rocchetti (vangieson@libero.it)
Ernesto Mezza (mezza@unina.it)
Mario Nasti (mario.nasti@tin.it)
Viviana Strazzullo (graziafanara@libero.it)
Vittorino Montanaro (vincenzo.santoro0@alice.it)
Massimo Mascolo (mmascolo@libero.it)
Pantaleo Bufo (p.bufo@unifg.it)

Version: 5

Date: 17 December 2006

Author's response to the Biomed Central Editorial Production Team:

Thank you for consideration of our manuscript for publication in your journal.
We are very grateful for your further revisions and for your lots of useful advice.
By this cover letter, we want to provide point-by-point responses to your comments.
Our paper has been written by observing and respecting the journal style.
We have tried to obtain a new revised and correctly formatted manuscript.
In light of your comments, below, you can read a detailed description of all changes we have made since the previous version.

Formatting changes requested

-------------------------------------
Title - Please remove the full stop from the end of the title.
Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names. Can we ask you to link each author to their corresponding affiliation with the use of superscript numbers.

Running title - Please remove the running title from the title page.

Keywords section - Please remove the keywords section from the manuscript.

References - The reference list should contain all authors names, if there are 30 or below. The term 'et al.' should not be used. We have modified and corrected the references in accordance with your instructions.

Acknowledgements - Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. We have specified that the source of funding for the publication of our scientific work becomes from the Department of Surgical Sciences, University of Foggia, Foggia, Italy.

Tables - We are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. We have eliminated shading and colours within the tables.

Table title and legend - The table title should appear above the table. The legend (or key) should appear below the table.

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. The figures have been cropped.

We have carefully checked our manuscript.

We have agreed to pay the processing charge. Please remember that, when you decide on publication of our manuscript, a pro-forma invoice (as fax) can be addressed to:

Dr. Pio D'Errico
Fax: 0881 713032
Department of Surgical Sciences
University of Foggia
Foggia (ITALY)

With our best wishes of Merry Christmas and Happy New Year,

Dr. Giuseppe Pannone